RESEARCH ARTICLE

Adverse fetal and neonatal outcomes in pregnancies with confirmed Zika Virus infection in Rio de Janeiro, Brazil: A cohort study

Juliana P. Souza¹, Maria Dalva B. B. Méio¹, Laura Medeiros de Andrade¹, Mirza R. Figueiredo¹, Saint Clair Gomes Junior⁰¹, Jose Paulo Pereira Junior¹, Elizabeth Brickley², Maria Elisabeth Lopes Moreira⁰¹*

1 Instituto Nacional de Saúde da Mulher da Criança e do Adolescente Fernandes Figueira–IFF/Fundação Oswaldo Cruz–Fiocruz, Rio de Janeiro, Rio de Janeiro, Brazil, 2 Assistant Professor of Infectious Disease Epidemiology, Faculty of Epidemiology and Population, London School Hygiene and Tropical Medicine, London, United Kingdom

* bebethiff@gmail.com

Abstract

Objective

To analyze adverse fetal and neonatal outcomes of Zika virus infection by the timing of infection during pregnancy. Method: Cohort study of 190 pregnancies with 193 offspring with a positive RT-PCR test for Zika virus (March/2016 to April/2017).

Results

Death or defects related to congenital Zika virus infection were identified in 37.3% of fetuses and newborns, and microcephaly in 21.4% of the newborns. The proportion of small for gestational age newborns was 21.9%. Maternal symptoms in the first trimester were significantly associated with the birth of newborns with microcephaly/cerebral atrophy, small for gestational age and with the deaths (one abortion, one stillbirth and the two neonatal deaths). Maternal infection during the second trimester was further associated with asymptomatic newborns at birth. The study showed that 58.5% of the offspring with microcephaly and / or cortical atrophy were small for gestational age, with an evident decrease in symptomatic offspring without microcephaly, 24.1%, and with only 9.1% in the asymptomatic group.

Conclusion

This study showed that the earlier the symptoms appear during gestation, the more severe the endpoints. We found a higher percentage of small for gestational age newborns exposed to Zika virus early in gestation. We also found a group of apparently asymptomatic newborns with proven Zika infection, which highlights the importance of follow up studies in this population.



OPEN ACCESS

Citation: Souza JP, Méio MDBB, de Andrade LM, Figueiredo MR, Gomes Junior SC, Pereira Junior JP, et al. (2021) Adverse fetal and neonatal outcomes in pregnancies with confirmed Zika Virus infection in Rio de Janeiro, Brazil: A cohort study. PLoS Negl Trop Dis 15(1): e0008893. https://doi. org/10.1371/journal.pntd.0008893

Editor: Marcus Vinícius Guimarães Lacerda, Fundacao Oswaldo Cruz, BRAZIL

Received: December 23, 2019

Accepted: October 16, 2020

Published: January 4, 2021

Copyright: © 2021 Souza et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because public availability would compromise patient privacy. Access to deidentified data can be made available upon reasonable request from qualified investigators by contacting heal@lshtm.ac.uk.

Funding: The study was supported by Fiocruz PIP/ IFF program (to MELM); CNPq 441098/2016-9 and 305090/2016-0 (to MELM); Faperj E_18/2015TXB (to MELM); Wellcome Trust & the United Kingdom's Department for International Development (205377/Z/16/Z) (to EB) and European Union's Horizon 2020 research and innovation programme under ZikaPLAN grant agreement no.734584 (to EB). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Zika virus infected pregnant women brought many challenges for the public health system because at the beginning of the outbreak there was no knowledge of the consequences of this disease. Microcephaly became the more common and threatening finding related to intrauterine Zika virus infection. Moreover, it was observed that there were others serious consequences of this disease during pregnancy, which were a burden for mothers and children. This study, following infected pregnant women, referred to our institution, enabled the medical professional team to improve their expertise in dealing with this new disease. Ours results showed a high frequency of intrauterine growth restriction, with babies being born small. Besides, it was found that the more severe cases occurred in women who were infected during the first trimester of gestation. Infections during the last half of the gestation more frequently leaded to babies born without symptoms. However, these results highlight the importance of following the babies born apparently asssymptomatic.

Introduction

Congenital Zika virus (ZIKV) infections have the potential to severely affect fetal development and can lead to a wide range of adverse pregnancy and neonatal outcomes, including spontaneous abortion, fetal loss, low birth weight, and congenital malformations, especially in the central nervous system [1]. Despite some other studies involving pregnant women with confirmed ZIKV infection [2–6] gaps in knowledge are still found, especially regarding the broad spectrum of perinatal involvement of congenital ZIKV infection, ranging from apparently asymptomatic newborns to fetal and neonatal death.

Although the different methods used in the evaluation and classification of the newborn may have contributed to the differing risk estimates, it remains uncertain whether other biological or environmental co-factors could contribute to the observed heterogeneity [7]. Several studies to date have provided evidence that higher risks of adverse outcomes are associated with early, as compared to later, ZIKV infections during pregnancy [2,4,5].

Based on these findings, we hypothesized that earlier maternal infections may cause more deleterious embryonic or fetal compromise and, therefore, more severe repercussions for prenatal development. Thus, this study aimed to evaluate the adverse fetal and neonatal endpoints due to ZIKV congenital infection by timing of viral infection during gestation and to describe the range of abnormalities observed among the children with Congenital Zika Syndrome.

Materials and methods

Ethics statement

The study followed regulatory standards and guidelines established under Resolution N° 466/ 2012 of the National Health Council and demands from the IFF Research Ethics Committee. The subscribed number at CAAE 526756616000005269 referred to the Project "Vertical Exposure to Zika Virus and Its Consequences for Child Neurodevelopment" and registered in Clinical Trials.gov Identifier: NCT03255369. Written consent forms were obtained from the pregnant women before inclusion in the study.

Study design

A cross sectional study was performed in a cohort of pregnant women exposed to ZIKV infection analyzing the perinatal repercussions in pregnancy considering the trimester of exposition (NCT03255369).

Study site. The study was conducted at the Obstetrics and Neonatology Services of the Fernandes Figueira National Institute of Women, Children and Adolescent Health/Oswaldo Cruz Foundation (IFF/FIOCRUZ), a public tertiary hospital located in Rio de Janeiro, Brazil. This hospital is a reference center for congenital malformations and has provided specialized care for the ZIKV-infected women and their children in the city of Rio de Janeiro since the onset of 2015–2017 epidemic.

Population

The study population was recruited from a convenience sample of 194 pregnant women who were referred by public and private healthcare services after laboratory diagnosis, clinical symptoms, or evidence of fetal alteration during gestational ultrasound suggestive of ZIKV infection.

Pregnant women with acute febrile syndrome, cutaneous rash or fetal ultrasound with malformations suspected for Zika syndrome who tested positive by RT-PCR (urine or blood) were included in the study, as well newborn infants with some suspected malformations related do ZIKV who tested positive by RT-PCR.

Exclusion criteria

The exclusion criteria were specific situations found in the study's population unrelated to ZIKV congenital infection. One case of type II Arnold Chiari was excluded, as there was no certainty if the ZIKV infection occurred before or after the diagnostic of the malformation. Newborns with chromosomic alterations detected during fetal life by amniocentesis or at birth, were excluded from the original cohort. Also, all fetuses and newborns with laboratory-confirmed congenital infection by other TORCH [8] group agents were excluded at the admission in the original cohort.

Exposure

The primary exposure was the trimester of the proven maternal ZIKV infection. Gestational trimesters were defined as: first trimester: up to 13 weeks and 6 days; second trimester: from 14 to 27 weeks and 6 days; and third trimester: after 28 weeks and until 40 weeks and 6 days [9]. The timing of maternal infection during gestation was based on the reporting of the first signs and symptoms suggestive of ZIKV infection, as per the manifestations described by the Brazilian Ministry of Health [10], with the typical rash from the disease appearing as one of the most striking signs.

Variables

The analysis evaluated both maternal, perinatal, anthropometric, and clinical variables. The maternal variables were related to both socioeconomic conditions and gestational information, such as the numbers of pregnancies and prior spontaneous abortions, maternal hypertensive diseases, premature labor, and premature rupture of membranes. The perinatal variables included the date of birth, date of fetal or neonatal death, delivery type (i.e., vaginal or C-section), APGAR score (i.e., 1st and 5th minute), gestational age (i.e., as determined by last menstrual period, obstetric ultrasound before 12 weeks and the newborn's physical examination,

New Ballard Score [11], in this order [12]). Gestational age was analyzed in whole weeks, without specification of days, as recommended by the WHO [13]. The anthropometric variables included birth weight (BW) in grams (g) and head circumference (HC) in centimeters (cm) with the respective Z-scores to age and gender as per the InterGrowth21 neonatal growth chart [14].

The clinical variables were related to examinations and evaluations of the newborns that occurred after birth and up to 28 days of life, and included neuroimaging, ophthalmic, neurologic, and orthopedic exams. Imaging exams included transfontanellar doppler ultrasound in all newborns, cranial computed tomography scan and cranial magnetic resonance imaging (MRIs) with or without contrast in cases of suspected alterations, with reports provided by pediatric radiologists. The ophthalmic evaluations included indirect ophthalmoscopy conducted by pediatric ophthalmologists with experience in congenital infection in the neonatal population. Neurological evaluations were conducted by neuropediatricians and general clinical evaluation by pediatric infectious disease specialists. Orthopedic evaluations were performed by the pediatric orthopedics service when necessary following clinical referral.

Outcomes

The primary fetal and neonatal outcomes considered in this study were based on the previously stated patterns reported in other studies, and the most up-to-date information about mother-to-child ZIKV infection [15]. All outcomes were defined when pregnancy ended or within the first 28 days of life, regardless of gestational age.

The primary outcomes included:

- 1. Abortion defined as the expulsion or extraction of a product of the conception with less than 500g and /or height less than 25 cm, or less than 22 weeks of gestation, with or without evidence of life [16].
- 2. Fetal death (Stillbirth) defined as the death of the gestational product before expulsion or its complete extraction from the mother's body, regardless of pregnancy duration [16].
- 3. Neonatal death defined as a death occurring between 0 and 28 days of birth [17].
- 4. Small for gestational age (SGA) birth defined as a birth with birthweight below the 10th percentile (or Z score < minus 1.28) as per the InterGrowth 21neonatal growth chart [14]. We have also considered the cutoff point of birth weight below the third percentile (or Z-score < minus 1.88), as evidence of more severe intrauterine growth restriction, according to the definition of Lee et al. (2003) [18].</p>
- 5. Preterm birth defined as a birth with a gestational age of less than 37 weeks of pregnancy.
- 6. Symptomatic with microcephaly and/or cerebral atrophy defined as children with microcephaly and/or cerebral atrophy identified through clinical and neuro-imaging evaluations. Microcephaly was assessed in the first 24–48 hours of life and defined, respectively, as moderate or severe with a HC of < minus 2 to minus 3 or of < minus 3 standard deviations below the mean for sex and age as per the InterGrowth21 neonatal growth chart [19];</p>
- Symptomatic without microcephaly or cerebral atrophy defined as children with other signs and symptoms consistent with congenital ZIKV infection as described by other authors [15] and identified through the neuroimaging, ophthalmic, neurologic, and orthopedic exams;
- 8. Asymptomatic defined as children with normal morphological and functional evaluations within the first 28 days of life.

Statistical analysis

The statistical analysis was performed with the program EPI-INFO version 7.2; all p-values were calculated for 2-sided statistical tests using a 0.05 level of significance. Descriptive analysis was conducted to estimate frequencies of adverse outcomes. Continuous variables were summarized using mean and standard deviations or median and interquartile range, depending on the distribution of the data and analysis of proportions to discrete or categorical variables. The prevalence of adverse outcomes was estimated by the gestational period in which the viral infection occurred. Chi-squared tests were used to compare proportions between categories; Fisher's exact tests were used for variables with cells including 5 or fewer observations.

Results

Of the 191 pregnancies, 194 offspring had positive RT-PCR results for ZIKV in any studied biological material (Fig 1). After one exclusion, a newborn with Arnold Chiari Syndrome, the total study population, which included three twin pregnancies, consisted of a total of 193 exposed fetuses and newborns.

Overall, the participating women had a mean age of 29.5 years (SD = 6.8), varying from 14 to 45 years. Most of these women had attained at least high school (63.1%), 55% self-reported as white, and the average monthly Family income (Brazilian Reais) was 3,400 (SD = 3,840). The primary perinatal data of the studied population are shown in Table 1. There were few comorbidities, such as maternal hypertension (13,3%) and gestational diabetes (3,9%). Among the neonates, 69,8% were delivered by Caesarean section, and the majority (84.3%) were delivered at term with a mean gestational age of 38.2 weeks (Table 1).

Adverse fetal and neonatal outcomes were identified in 72 offsprings (37.3%) of the 193 exposed fetus, with one abortion and one stillbirth. There were four neonatal deaths until 28 days of life; in all four cases, the mothers had a viral infection in the first gestational trimester. From the 193 exposed fetus, 70 (36.6%) presented with abnormalities upon examination and 121 offspring, representing 63.3%, were asymptomatic upon evaluation. Among the 70 symptomatic neonates at birth, 41 had microcephaly and/or cerebral atrophy (58.6%), and 29 neonates (41.4%) had other characteristics suggestive of ZIKV congenital disease, as described by Moore et al [15], without microcephaly or cerebral atrophy. (Fig 2). Among the neonatal deaths, three were born with evidence of microcephaly and/or cerebral atrophy and one only presented with the characteristics of congenital ZIKV infection but without microcephaly. All four neonatal deaths were considered small for gestational age, and one was delivered prematurely. Microcephaly in the full study population totaled 36 cases (18.7%), of which 27 cases (14.1%) were classified as severe microcephaly.

Out of the 184 pregnancies with known period of maternal ZIKV symptoms, 64 (34.8%), 85 (46.2%), and 35 (19.0%) women reported the symptoms in the first, second, and third trimesters of pregnancy, respectively. In nine cases, the mothers could not report the period of infection. The symptomatic newborns, with and without microcephaly, were exposed predominantly in the first trimester, while those born asymptomatic were exposed mainly in the second trimester (Table 2). Also, the SGA newborns were predominantly exposed in the first trimester. In this population, 29 newborns were preterm (15.0%), of which 26 (89,6%) were classified as moderate preterm (i.e., between 32 weeks and 36 weeks and 6 days of gestational age). There was no significant difference in the trimester of gestation exposition for premature birth (Table 2).

In the study population, 42 of 193 deliveries (21.9%) were SGA (i.e., less than the tenth percentile). There were 12 offspring who were both SGA and premature at delivery (i.e., 6.2% of deliveries overall). The distribution of SGA cases also varied by the phenotype of liveborn



* 2 twin pregnancies; one twin was excluded due to congenital syphilis.

RT-PCR - Reverse transcription polymerase chain reaction; IVH - neonatal intraventricular hemorrhage.

Fig 1. Strobe Flowchart for selection of study subjects. There are three twin pregnancies in the sample.

https://doi.org/10.1371/journal.pntd.0008893.g001

neonates, with of 31/42 SGA cases (73.8%) exhibiting symptoms consistent with congenital ZIKV infection (Fig 2). Considering a more restrictive cutoff point to define SGA, 38 (90.4%) of the 42 SGA deliveries fell below the third percentile. In this sample, 58.5% of the cases classified in Group A (i.e., symptomatic with microcephaly/cerebral atrophy) and 24.1% classified in Group B (i.e., symptomatic without microcephaly/cerebral atrophy) were SGA (Table 3). Notably, there were 23 SGA under the third percentile in Group A.

VARIABLE	N	MEAN (SD)	N (%)
Multiple gestation	193		3 (1.5)
Maternal hypertension	181		24 (13.3)
Maternal diabetes	180		7 (3.9)
Maternal TORCH coinfection	193		6 (3.1)
Preterm birth	193		30 (15.5)
Male sex	191		95 (49.7)
Cesarean section	189		132 (69.8)
Gestational age at delivery (weeks)	193	38.2 (2.9)	
HC at delivery (cm)	189	33.3 (3.2)	
HC at delivery (Z-score)	189	0.17 (2.5)	
Birthweight (g)	192	2976.14 (709.3)	

Table 1. Demographic and perinatal characteristics of the offspring (N = 193).

TORCH—Toxoplasma gondii, others (Listeria monocytogenes, treponema pallidum, varicella-zoster virus, human immunodeficiency virus, enterovirus and parvovirus B19), rubella virus, cytomegalovirus, and herpes simplex virus. HC-head circumference.

The diferences in the sample number is due to missing values. There was one abortion from whom there weren't information on sex and anthropometry; and one stillbirth, accounting for the lack of information on sex.

https://doi.org/10.1371/journal.pntd.0008893.t001

The clinical presentation of neonates with symptoms consistent with congenital ZIKV infection were significantly associated with the first trimester of exposition to ZIKV. The most severe abnormalities were predominantly related to the first trimester infection, although the exposition in the second trimester was also associated with the impairment of fetus development (Table 4). Microphtalmia and Coloboma were not related to the trimester of infection, though the numbers of affected children were very small (Table 4).

Discussion

This study recruited the pregnant women during the outbreak of ZIKV infection which occurred in Brazil between 2015-2017. This study showed that the earlier the symptoms appear during gestation, the more severe the endpoints. About half of the patients complained of viral infection signs and symptoms during the second trimester, which is in line with what it was reported by Van Maldeghem K et al. [5] and França V A et al [20,1]. Within adverse pregnancy endpoints, neonates that were symptomatic with microcephaly/cerebral atrophy, small for gestational age (SGA) and stillborn were more common in women that had symptoms earlier during their pregnancy, which confirms data previously described (3-5) [21]. Microcephaly was present in 21.2% of the offsprings, which is significantly higher when compared to the systematic review of Campos Coelho A. V. et al. (2017) [21], which reported a range from 1 to 13%, depending on location, to the 1% for French Polynesia [22] and to the 13% for the state of Bahia [23]. However, our institution is a reference for malformations and received pregnant women or newborn after birth with malformations and this can explain high rate of microcephaly. Previous results from the same cohort in Rio de Janeiro, including only the pregnant women with symptoms (fever and cutaneous rash) and positive RT-PCR results, published recently, described only four cases of microcephaly (3.4%) (2). Once again, most microcephaly cases were related to maternal viral infection during the first trimester, as occurred in our sample. Nonetheless, our high incidence of microcephaly could be explained because the hospital is a referral center for congenital abnormalities during pregnancy. Our study showed no case of microcephaly associated with maternal viral infection during the third trimester.



Group A: Symptomatic newborn with microcephaly / cerebral atrophy; **Group B:** Symptomatic newborn without microcephaly / cerebral atrophy; **Group C:** Asymptomatic newborn

* 2 twin pregnancies, one twin was excluded due to congenital syphilis SGA- Small for gestational age (<10th percentile according to the InterGrowth21 Neonatal growth chart); Preterm (<37 weeks of gestational age).

Fig 2. Flowchart of study outcomes (N = 191^a). ^a Excluding one miscarriage and one stillbirth. SGA- Small for gestational age ($<10^{th}$ percentile according to the InterGrowth21 Neonatal growth chart); Preterm (<37 weeks of gestational age). SGA and preterm categories are not mutually exclusive. Group A: Symptomatic with microcephaly and/or cerebral atrophy; Group B: Symptomatic without microcephaly or cerebral atrophy; Group C: Asymptomatic.

https://doi.org/10.1371/journal.pntd.0008893.g002

Our research found that 37.3% of pregnancies with exposition to ZIKV resulted in fetal or neonatal adverse outcomes. In the literature, these proportions range from 5% [5] to 6% [4] in two North American studies, 7% in French American territory (3), 55% in the French Guiana [6]. A case control study in the Northeast of Brazil confirmed the association of microcephaly with ZIKV exposure during gestation and refers 43% of association of cerebral abnormalities with ZIKV infection in positive cases [24] The studies cited employed different methodologies, related to the availability of imaging tests, selection of the exposed, and evaluation of the newborns, which can explain the variation of the percentages found, much lower than our study. One of the explanations for this finding could be related to the more detailed investigation of the newborns in our study, with more sophisticated imaging tests performed in every altered

		GESTATIONAL	TRIMESTER N (%)		
OUTCOME	N	1st	2nd	3rd	p-value ^a
GROUP A	35	29 (82.9)	6 (17.1)	0	<0.0001
GROUP B	28	13 (46.4)	10 (35.7)	5 (17.9)	<0.0001
GROUP C	119	21 (17.6)	68 (57.1)	30 (25.2)	<0.0001
SGA	38	21 (33.3)	15 (17.6)	2 (5.7)	0.003
Premature	28	10 (15.9)	13 (15.3)	5 (14.3)	0.978

Table 2. Relationship between the trimester of infection and the clinical outcomes ($N = 182^{b}$).

a Pearson Chi Square was used to compare the distribution of the groups, the prevalence of SGA and the prevalence of premature in relation of the trimester of gestation.

SGA- Small for gestational age (<10th percentile on InterGrowth21 Neonatal growth chart).

Group A: Symptomatic with microcephaly and/or cerebral atrophy; Group B: Symptomatic without microcephaly or cerebral atrophy; Group C: Asymptomatic. **b** Exclusion from the analysis: nine cases with uncertain trimester of exposition, one abortion and one stillbirth in which the presence of microcephaly was not possible to confirm.

https://doi.org/10.1371/journal.pntd.0008893.t002

case, with cranial MRIs and CT scans, as well as multi-professional evaluation until 28 days of age. Another explanation is that this was a reference hospital for ZIKV infection in pregnant women and newborns, and also in fetal malformations in general.

The proportion of prematurity in this study was similar to the world percentage, about 11% according to WHO [12], and 11.5% in Brazil according to Birth in Brazil research [25], as well as in the population exposed to ZIKV: 13% in French Guiana [6] and 8.8% in New York City [26]. Therefore, it could be assumed that mother-to-child ZIKV infection does not lead to higher levels of premature births.

We found a higher percentage of small for gestational age newborns (21.9%) compared to the Brazilian population and to other studies concerning the newborn population exposed to ZIKV during gestation [3,26,27,28,29]. The study by Lee et al. (2013), including the population of several underdeveloped and developing countries, reported a prevalence of 12.5% in Latin America and the Caribbean [27], while in the population exposed to ZIKV, the proportions ranged from 9% [2] to 13% [3], using the same definitions proposed by this study. The comparison of the prevalence of SGA newborns in Brazil is limited since national studies include different profiles of hospital complexity, specific populations from each region and the use of diverse methods for the classification of SGA newborns. In our study, the proportion of SGA newborns was as significant as the more severe involvement of newborns, which is in agreement with the study by Prata-Barbosa et al. [28]. The same trend occurred concerning the distribution throughout gestational trimesters: the earlier the maternal infection, the higher the percentage of SGA newborns. Others studies with ZIKV-infected mothers, considering the same definition for IUGR, reported a prevalence of SGA of 11.2% in Cooper et al [26], 12% in

Table 3. Association of the offspring congenital Zika virus infection symptomatology with frequency of SGA, also considering a more restrictive cutoff point	
(N = 191).	

	GROUP A n (%)	GROUP B n (%)	GROUP C p-value n (%)	
SGA < p10	24 (58.5)	7 (24.1)	11 (9.1)	<0.0001
SGA < p3	23 (59.0)	5 (17.9)	10 (8.3)	<0.0001

SGA-Small for gestational age, below the 10th (P10) and 3rd percentile (P3), respectively, on InterGrowth21 Neonatal growth chart.

Group A: Symptomatic with microcephaly and/or cerebral atrophy; Group B: Symptomatic without microcephaly or cerebral atrophy; Group C: Asymptomatic.

https://doi.org/10.1371/journal.pntd.0008893.t003

CHARACTERISTICS	N	1 st TRIMESTER n (%)	2 nd TRIMESTER n (%)	3rd TRIMESTER n (%)	p-value
CRANIAL MORPHOLOGY					
Microcephaly	182	24 (38.1)	6 (7.1)	0	< 0.0001
BRAIN ANOMALIES					
Cerebral cortex thinning	171	27 (46.6)	6 (7.3)	0	< 0.0001
Abnormal gyral patterns	171	26 (44.8)	6 (7.3)	0	< 0.0001
Corpus callosum anomalies	171	8 (13.8)	5 (6.1)	0	0.05
Cerebellar hypoplasia	171	8 (13.8)	1 (1.2)	0	0.002
Decreased white matter	171	11 (20.4)	4 (4.9)	0	0.001
Ventriculomegaly	170	29 (37.9)	3 (3.7)	0	< 0.0001
Subcortical calcification	171	32 (55.2)	5 (6.1)	0	< 0.0001
MUSKULOSKELETAL ANOMALIES					
Congenital clubfoot	179	3 (4.9)	0	0	0.052
Arthrogryposis	179	6 (9.8)	2 (2.4)	0	0.038
Hip dislocation	179	5 (8.2)	1 (1.2)	0	0.033
NEUROLOGICAL IMPAIRMENT					
Motor disability	175	31 (53.4)	10 (12.0)	1 (2.9)	< 0.0001
Hypertonia	175	32 (55.2)	9 (10.8)	3 (8.8)	< 0.0001
Pyramidal syndrome	175	29 (50.0)	7 (8.4)	2 (5.9)	< 0.0001
Hyperexcitability	175	21 (36.2)	6 (7.2)	0	< 0.0001
Epilepsy	175	27 (46.6)	6 (7.2)	0	< 0.0001
Swallowing dysfunction	175	13 (22.4)	1 (1.2)	1 (2.9)	< 0.0001
OPHTALMOLOGICAL ANOMALIES					
Microphthalmia	173	1 (1.8)	0	1 (2.9)	0.352
Coloboma	173	1 (1.8)	2 (2.4)	1 (2.9)	0.931
Chorioretinal atrophy	173	16 (28.1)	4 (4.9)	0	< 0.0001
Pigment mottling	173	12 (21.1)	1 (1.2)	0	< 0.0001
Optic nerve hypoplasia	173	15 (26.3)	2 (2.4)	0	0.0001

Table 4. Common anomalies associated with congenital Zika Virus infection in this sample of symptomatic offspring per trimester of gestation (N = 184).

Pearson Chi Square test.

11 offsprings were excluded: 9 with uncertain period of infection during gestation and 2 fetal deaths (one miscarriage and one stillbirth due to the impossibility of clinical investigation.

In 14 newborns there were muskuloskeleton anomalies. In 49 there were neurological abnormalities and 173 had an ophtalmological evaluation.

List of features modified from Moore et al-Characterizing the pattern of anomalies in Congenital Zika Syndrome for pediatric clinicians. JAMA Pediatric.2017;171 (3):288–295 (15).

https://doi.org/10.1371/journal.pntd.0008893.t004

French Guiana [6] and 18% in Walker CL et al. [29], who also found head and abdominal circumferences below the 10th percentile in the ultrasounds performed. Thus, it could be inferred that symmetrical IUGR in fetal ultrasounds might be an early gestational marker of a newborn probably vertically infected by ZIKV.

The referred population to this facility during the Zika epidemic originated from both private and public services, which could justify the higher maternal income and education level of the studied population.

The strengths and limitations of the study warrant consideration. The key strength of the study is that affected newborns were evaluated through specialized clinical investigations and accurate imaging tests of the central nervous system. Cranial CT scans and MRIs were made in almost every affected neonate, which also contributed to the identification of a more significant number of the alterations found. One limitation of this study was that the inclusion of

some women after the fetal ultrasound with mycrocephaly. These fetal outcomes were detected during the routine ultrasound in asymptomatic women and they were included in the recruitment process, when previous fetal development impairment could not be known for certain, so possibly introducing a selection bias. However, this was a reference hospital for ZIKV infections in pregnant women during an epidemic, and they were referred because, although not having symptoms of ZIKV infection, they were tested after a fetal ultrasound showing abnormalities, and they had a positive RT-PCR in any biological material. Another limitation was the size of the sample, which did not allow us to evaluate the occurrence of SGA adjusting for others causes, for example, maternal hypertension, as we had only 13.3% of them in this sample. We are also not describing a control group in this paper.

Conclusions

The study confirmed the already reported relationships in other papers, which is the earlier the ZIKV infection during pregnancy, the more severe the fetal endpoints and, therefore, the greater morbimortality in the affected subjects. The higher percentage of SGA newborns found in this study deserves attention, not only for the adequate management of the affected population during the epidemics, but also the investigation of its causal mechanisms. This study also identified a group of neonates without microcephaly but who presented with other anomalies, neurological or clinical, and furthermore, we also identified another group that although having a proven exposition during pregnancy were born apparently normal. This is a matter of concern, as one cannot predict if these babies will remain with a normal development in future. A long-term longitudinal study must be considered to evaluate better the endpoints brought by this congenital infection, especially the asymptomatic newborns, which could help the understanding of the long-term consequences of this infection in pregnant women.

Author Contributions

- **Conceptualization:** Maria Dalva B. B. Méio, Jose Paulo Pereira Junior, Elizabeth Brickley, Maria Elisabeth Lopes Moreira.
- **Data curation:** Juliana P. Souza, Maria Dalva B. B. Méio, Laura Medeiros de Andrade, Elizabeth Brickley.
- Formal analysis: Juliana P. Souza, Maria Dalva B. B. Méio, Laura Medeiros de Andrade, Saint Clair Gomes Junior.

Funding acquisition: Maria Elisabeth Lopes Moreira.

Investigation: Juliana P. Souza, Jose Paulo Pereira Junior.

- Methodology: Juliana P. Souza, Maria Dalva B. B. Méio, Mirza R. Figueiredo, Elizabeth Brickley, Maria Elisabeth Lopes Moreira.
- Project administration: Maria Elisabeth Lopes Moreira.

Resources: Elizabeth Brickley, Maria Elisabeth Lopes Moreira.

Software: Saint Clair Gomes Junior.

Supervision: Maria Dalva B. B. Méio.

Validation: Saint Clair Gomes Junior.

Writing – original draft: Juliana P. Souza, Maria Dalva B. B. Méio, Laura Medeiros de Andrade, Mirza R. Figueiredo, Jose Paulo Pereira Junior, Elizabeth Brickley.

Writing – review & editing: Juliana P. Souza, Laura Medeiros de Andrade, Mirza R. Figueiredo, Jose Paulo Pereira Junior, Elizabeth Brickley, Maria Elisabeth Lopes Moreira.

References

- Carvalho-sauer R, Nascimento C, Barreto FR, Gloria M. International Journal of Infectious Diseases Congenital Zika Syndrome: Prevalence of low birth weight and associated factors. Bahia, 2015–2017. Int J Infect Dis [Internet]. 2019; 82:44–50. Available from: <u>https://doi.org/10.1016/j.ijid.2019.02.040</u> PMID: 30831221
- 2. Brasil P, Moreira ME, Nogueira RMR, Damasceno L, Wakimoto M, Rabello RS, et al. Zika Virus Infection in Pregnant Women in Rio de Janeiro. N Engl J Med. 2017; 375(24):2321–34.
- Hoen B, Schaub B, Funk AL, Ardillon V, Boullard M, Cabié A, et al. Pregnancy Outcomes after ZIKV Infection in French Territories in the Americas. N Engl J Med [Internet]. 2018; 378(11):985–94. Available from: http://www.nejm.org/doi/10.1056/NEJMoa1709481 PMID: 29539287
- Honein MA, Dawson AL, Petersen EE, Jones AM, Lee EH, Yazdy MM, et al. Birth defects among fetuses and infants of US women with evidence of possible zika virus infection during pregnancy. JAMA —J Am Med Assoc. 2017; 317(1):59–68. https://doi.org/10.1001/jama.2016.19006 PMID: 27960197
- Territories US, VanMaldeghem K, Prado MV, Ellis E, Anesi MS, Simeone RM, et al. Pregnancy Outcomes After Maternal Zika Virus Infection During Pregnancy. MMWR Morb Mortal Wkly Rep. 2017; 66 (23):615–21. https://doi.org/10.15585/mmwr.mm6623e1 PMID: 28617773
- 6. Pomar L, Vouga M, Lambert V, Pomar C, Hcini N, Jolivet A, et al. Maternal-fetal transmission and adverse perinatal outcomes in pregnant women infected with Zika virus: Prospective cohort study in French Guiana. BMJ. 2018;363. https://doi.org/10.1136/bmj.k4431 PMID: 30381296
- Lee H, Lotery A. Risk of Zika-related microcephaly: stable or variable? 2017; 390:824–6. https://doi.org/ 10.1016/S0140-6736(17)31478-2 PMID: 28647174
- Neu N, Duchon J, Zachariah P. TORCH infections. Clin Perinatol [Internet]. 2015; 42(1):77–103. Available from: https://doi.org/10.1016/j.clp.2014.11.001 PMID: 25677998
- Pettker CM, Goldberg JD, El-Sayed YY. Methods for Estimating the Due Date Committee on Obstetric Practice American Institute of Ultrasound in Medicine Society for Maternal–Fetal Medicine. 2017;(700). Available from: https://www.acog.org/-/media/Committee-Opinions/Committee-on-Obstetric-Practice/ co700.pdf?dmc=1&ts=20170719T0337358930
- Brasil Ministério da Saúde. Secretaria de Vigilância em Saúde. Departamento de Vigilância das Doenças Transmissíveis. Protocolo de vigilância e resposta à ocorrência de microcefalia relacionada à infecção pelo vírus Zika. Brasília: Ministério da Saúde, 2015 (in Portuguese).
- Ballard JL, Khoury JC, Wedig K, Wang L, Eilers-Walsman BL, Lipp R. New Ballard Score, expanded to include extremely premature infants. J Pediatr. 1991; 119(3):417–23. https://doi.org/10.1016/s0022-3476(05)82056-6 PMID: 1880657
- Brasil. Ministério Da Saúde. Atenção ao pré-natal de baixo risco [Internet]. Editora do Ministério da Saúde, 2012. 2012. 318 p. Available from: http://bvsms.saude.gov.br/bvs/publicacoes/cadernos_ atencao_basica_32_prenatal.pdf.
- 13. WHO recommendations on interventions to improve preterm birth outcomes. Available from: www.who. int/reproductivehealth.
- 14. InterGrowth21st. Oxford: Oxford University, 2016.: http://intergrowth21.tghn.org/fetal-growth.
- Moore CA, Erin SJ, B. DW, Pessoa A, V. VC, Fonseca EB, et al. Congenital Zika Syndrome: Characterizing the Pattern of Anomalies for Pediatric Healthcare Providers. Jama Pediatr. 2017; 171(3):288–95. https://doi.org/10.1001/jamapediatrics.2016.3982 PMID: 27812690
- 16. ICD 10.: https://www.icd10data.com/ICD10CM/Codes/P00-P96/P9. Available from: https://www. icd10data.com/ICD10CM/Codes/P00-P96/P95-/P95.
- 17. Estimates G. Neonatal and Perinatal Mortality. Available from: https://apps.who.int/iris/handle/10665/ 43444.
- **18.** Lee PA, Chernausek SD, Hokken-koelega ACS. International Small for Gestational Age Advisory Board Consensus. Pediatrics. 2013; 111(1253).
- 19. Ministério da Saúde. Orientações integradas de vigilância e atenção à saúde no âmbito da Emergência de Saúde Pública de Importância Nacional MINISTÉRIO DA SAÚDE Secretaria de Vigilância em Saúde Secre1. DECRETO LEGISLATIVO No 395, DE 2009—Publicação Original—Portal Câmara. 2017;158. Available from: http://portalarquivos.saude.gov.br/images/pdf/2016/dezembro/12/ orientacoes-integradas-vigilancia-atencao.pdf.

- 20. França GVA, Schuler-Faccini L, Oliveira WK, Henriques CMP, Carmo EH, Pedi VD, et al. Congenital Zika virus syndrome in Brazil: a case series of the first 1501 livebirths with complete investigation. Lancet (London, England). 2016. https://doi.org/10.1016/S0140-6736(16)30902-3 PMID: 27372398
- **21.** Campos Coelho AV, Crovella S. Microcephaly prevalence in infants born to zika virus-infectedwomen: A systematic review and meta-analysis. Int J Mol Sci. 2017; 18(8):1–10.
- Cauchemez S, Besnard M, Bompard P, Dub T, Guillemette-Artur P, Eyrolle-Guignot D, et al. Association between Zika virus and microcephaly in French Polynesia, 2013–15: A retrospective study. Lancet. 2016. https://doi.org/10.1016/S0140-6736(16)00651-6 PMID: 26993883
- 23. Johansson MA, Mier-Y-Teran-Romero L, Reefhuis J, Gilboa SM, Hills SL. Zika and the Risk of Microcephaly.
- Velho T, Araújo B De, Arraes R, Ximenes DA, Miranda-filho DDB, Souza WV. Association between microcephaly, Zika virus infection, and other risk factors in Brazil: final report of a case-control study. 2018;18(March).
- 25. Lansky S, Friche AA de L, Silva AAM da, Campos D, Bittencourt SD de A, Carvalho ML de, et al. Pesquisa nascer no Brasil: perfil da mortalidade neonatal e avaliação da assistência à gestante e ao recém-nascido. Cad Saúde Pública [Internet]. 2014; 30(Sup):192–207. Available from: http://www. scielo.br/scielo.php?pid=S0102-311X2014001300024&script=sci_arttext
- Cooper HJ, Iwamoto M, Lash M, Conners EE, Paladini M, Slavinski S, et al. Maternal Zika Virus Infection. 2019; 134(6):1197–2.
- Lee ACC, Katz J, Blencowe H, Cousens S, Kozuki N, Vogel JP, et al. National and regional estimates of term and preterm babies born small for gestational age in 138 low-income and middle-income countries in 2010. Lancet Glob Heal. 2013; 1(1).
- Prata-Barbosa A, Martins MM, Guastavino AB, Cunha AJLA da. Effects of Zika infection on growth. J Pediatr (Rio J) [Internet]. 2018;(xx). Available from: https://linkinghub.elsevier.com/retrieve/pii/ S0021755718310271. https://doi.org/10.1016/j.jped.2018.10.016 PMID: 30593788
- Walker CL, Merriam AA, Ohuma EO, Dighe MK, Gale M, Rajagopal L, et al. Femur-sparing pattern of abnormal fetal growth in pregnant women from New York City after maternal Zika virus infection. Am J Obstet Gynecol. 2018; 219(2):187.e1-187.e20. <u>https://doi.org/10.1016/j.ajog.2018.04.047</u> PMID: 29738748